

Author's response to reviews

Title: Breastfeeding cessation and symptoms of anxiety and depression: a longitudinal cohort study

Authors:

Eivind Ystrom (eivind.ystrom@fhi.no)

Version: 5 **Date:** 10 May 2012

Author's response to reviews:

10 May 2012

Thank you for accepting my manuscript for publication in BMC Pregnancy and Childbirth. I have addressed the comments from the reviewer, and in the manuscript described how the study was granted and named the specific ethic committee that approved the study.

Sincerely,
Eivind Ystrom

Description of changes made to the manuscript

"Line 57 change 'is' to 'are risk factors'"

Thank you for noticing. This has now been corrected.

"Line 90 – Change except '2' to except 'two' hospitals."

Yes. The 2 has now been changed into 'two'.

"Line 101 – Change version 3 to version three"

This has been amended.

"Line 110-111 – Change present tense to past tense."

Thank you. This has now been changed.

"Line 114-115 – Please describe how missing data were managed in this study."

Yes, the ordering in this section was bewildering. Now the order is 1) the number of mothers with missing data, 2) how the missing data was distributed, 3) how the individuals with missing data was treated.